ŗ



Geophysical Research Letters

RESEARCH LETTER

10.1029/2021GL092958

Key Points:

- Dune height bistability drives alternating periods of barrier immobility and transgression over decades to centuries
- Simulations suggest discontinuous (continuous) retreat will become less (more) common with greater sealevel rise rates and storm intensity
- Dune dynamics are a critical driver of barrier evolution on time scales relevant to coastal management

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

I. R. B. Reeves, reevesi@live.unc.edu

Citation:

Reeves, I. R. B., Moore, L. J., Murray, A. B., Anarde, K. A., & Goldstein, E. B. (2021). Dune dynamics drive discontinuous barrier retreat. *Geophysical Research Letters*, *48*, e2021GL092958. https://doi. org/10.1029/2021GL092958

Received 12 FEB 2021 Accepted 7 JUN 2021

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

REEVES ET AL.

Dune Dynamics Drive Discontinuous Barrier Retreat

I. R. B. Reeves¹, L. J. Moore¹, A. B. Murray², K. A. Anarde¹, and E. B. Goldstein³

¹Department of Geological Sciences, University of North Carolina at Chapel Hill, Chapel Hill, NC, USA, ²Division of Earth and Ocean Sciences, Nicholas School of the Environment, Duke University, Durham, NC, USA, ³Department of Geography, Environment, and Sustainability, University of North Carolina at Greensboro, Greensboro, NC, USA

Abstract Barrier islands and spits tend to migrate landward in response to sea-level rise through the storm-driven process of overwash, but overwash flux depends on the height of the frontal dunes. Here, we explore this fundamental linkage between dune dynamics and barrier migration using the new model Barrier3D. Our experiments demonstrate that discontinuous barrier retreat is a prevalent behavior that can arise directly from the bistability of foredune height, occurring most likely when the storm return period and characteristic time scale of dune growth are of similar magnitudes. Under conditions of greater storm intensity, discontinuous retreat becomes the dominant behavior of barriers that were previously stable. Alternatively, higher rates of sea-level rise decrease the overall likelihood of discontinuous retreat in favor of continuous transgression. We find that internal dune dynamics, while previously neglected in exploratory barrier modeling, are an essential component of barrier evolution on time scales relevant to coastal management.

Plain Language Summary Barrier islands and spits tend to move landward in response to sea-level rise and storms when sediment from the beach is washed into the barrier interior and beyond, but tall dunes at the front of a barrier can prevent this process from happening for all but the largest storms. Here, we explore the interactions between dunes, storms, and barrier migration with a new model (Barrier3D). Our model shows that barriers migrate when dunes are low but are stationary when dunes are tall. Over decades to centuries, repeated cycles of dune loss and regrowth lead to sporadic (or "discontinuous") barrier retreat. Additionally, we find that sporadic behavior may become less common in the future in response to rising sea levels, while increasing storm intensity will change which environments are most likely to experience discontinuous retreat. These findings emphasize the importance of dune dynamics in controlling barrier evolution. Barrier migration over decadal timescales - timescales relevant to coastal communities and managers - is commonly modeled as a constant background process, but our results suggest that is not always the case.

1. Introduction

Coastal barrier systems (barrier islands and spits), which account for ~10% of the world's continental coastlines (Stutz & Pilkey, 2011), are of significant social, economic, and ecologic importance, yet the future of these landforms remains uncertain in the face of projected accelerated relative sea-level rise (RSLR; Sweet et al., 2017) and changes in tropical storm activity, including increasing intensity (e.g., Knutson et al., 2020). Barriers tend to migrate upward and landward in response to RSLR and storms, thereby maintaining subaerial exposure. Barrier migration occurs primarily through the process of overwash, where sediment eroded from the front of a barrier (shoreface and beach) is transported landward of the foredune crest to the barrier interior and back-barrier bay during storms (e.g., Dolan & Godfrey, 1973; Donnelly et al., 2006). Mounting evidence suggests that discontinuous (or "punctuated") barrier retreat, whereby a barrier oscillates between periods of migration and relative immobility over time scales of decades to centuries, is a prevalent behavior (Ashton & Lorenzo-Trueba, 2018; Ciarletta et al., 2019; Lorenzo-Trueba & Ashton, 2014; Mellett & Plater, 2018). This behavior contrasts with continuous transgression, where a barrier consistently retreats over time scales greater than the typical return period of overwash events. Sedimentological analyses of barriers and relict barrier deposits indicate that many barrier systems worldwide have experienced fluctuations in their rate of migration, which have been attributed to changes in external forcing such as the rate of RSLR (Cooper et al., 2016; Mellett et al., 2012); sediment supply (Forbes et al., 1991; Rampino & Sanders, 1982); shelf morphology and back-barrier accommodation (Nordfjord et al., 2009; Storms et al., 2008); antecedent

geology (Raff et al., 2018); and grain size (Forbes et al., 1995). Recently, experiments conducted using the exploratory model of Lorenzo-Trueba and Ashton (2014) suggest that decadal- to centurial-scale punctuated retreat can occur as the result of internal system dynamics, related specifically to lags in shoreface response to overwash and the assumption that overwash flux is related to a critical barrier width and height.

Here, we suggest that discontinuous retreat can also arise from dune dynamics, a suite of processes internal to barrier systems. Barrier dune systems tend to exist in one of two stable states as a result of feedbacks between dune recovery and storm erosion processes: a high-elevation, overwash-resistant state; or a low-elevation, overwash-vulnerable state (Durán Vinent & Moore, 2015; Durán Vinent et al., 2021; Goldstein & Moore, 2016). When biophysical dune recovery processes dominate, dunes tend to recover quickly if knocked down, remaining at or near their maximum height (Durán & Moore, 2013) and withstanding all but the largest of storms. When storm processes dominate, dunes cannot recover before the next storm and become trapped in a perpetual state of low elevation. However, when the time scales of dune recovery and storm occurrence are similar, foredune height becomes bistable (Durán Vinent & Moore, 2015; Durán Vinent et al., 2021; Goldstein & Moore, 2016). In this bistable regime, foredune systems respond nonlinearly to environmental drivers and are prone to abrupt shifts from one state to the other as the result of only small changes in environmental forcing. Under bistable conditions, the tendency for foredunes to exist in a state of either high or low elevation, and to alternate between the two states over time, has the potential to interact with barrier migration processes through alterations to overwash flux that occur with changes in dune height (Houser, Barrineau, et al., 2018; Sallenger, 2000). This fundamental linkage between dune dynamics and barrier evolution has yet to be explored using a long-term (decadal to millennial) modeling framework.

To investigate the role of dune dynamics in barrier migration, we introduce the new exploratory model Barrier3D, which builds upon earlier formulations for components of barrier evolution and incorporates previously neglected interactions between dunes and overwash. Previously, exploratory barrier models have coarsely resolved only large-scale, long-term changes in barrier position, height, and width and thus average over or neglect discrete events and associated temporal and alongshore variations in dune state (e.g., Lorenzo-Trueba & Ashton, 2014; Moore et al., 2010) in an effort to discover the processes most essential to barrier evolution. In contrast, Barrier3D tackles the scale separation between event-based and long-term models by explicitly, yet efficiently, simulating dune evolution, storm overwash, and a dynamically evolving shoreface in response to individual storm events and RSLR. Using this model, which we parameterize with data from Hog Island, VA, USA (a well-studied, undeveloped barrier island; Figure 1a), we explore barrier migration behaviors for a range of dune growth rates and storms frequencies. We then vary RSLR rates and storm intensities to explore how dune-storm interactions influence barrier migration under a range of anticipated future conditions.

2. Barrier3D

Barrier3D is an exploratory model (Murray, 2003, 2013) that resolves cross-shore and alongshore topographic variations to simulate the morphological evolution of a barrier segment over time scales of years to centuries. The model operates with a 1-year time step over a 10-by-10 m grid. The barrier segment consists of one or more alongshore rows of dune cells at the front (ocean side) of the grid, backed by an interior domain with a predetermined constant alongshore length and dynamically changing cross-shore width (Figure 1b). As explained below, dune and interior cells follow different sets of rules. Due to the complexities in modeling beach morphodynamics, and to focus on dune-storm interactions, Barrier3D does not simulate beach evolution and instead assumes invariant beach width and slope. Explanations of previously developed formulations and additional details not critical for understanding the fundamental workings of the model or its results are provided in the supporting information (Text S1) for completeness.

2.1. Storms and RSLR

RSLR is treated in the model using a Lagrangian frame of reference by reducing all elevations at the beginning of each modeled year relative to a sea level fixed at zero. Within each year, a probabilistically determined number of storms occur by randomly sampling from a normal distribution. We vary the mean of the distribution for our experiments and derive the standard deviation from a 35-year empirical storm record



Figure 1. Barrier3D model and parameterization. (a) Hog Island within the Virginia Coast Reserve (VCR), US mid-Atlantic coast, used for model parameterization and initial conditions. (b) Example Barrier3D model elevation grid with labeled domains. (c) Histogram of R_{high} values from 10,000 synthetic storms used in model runs. MSL is mean sea level.

for Hog Island developed using tide gauge and wave hindcast data. Each individual storm in Barrier3D is described by three statistics: (a) duration; (b) R_{high} , the highest elevation of the landward margin of runup; and (c) R_{low} , the lowest runup elevation. The statistics for each storm are determined by randomly selecting from a list of 10,000 synthetic storms (Figure 1c). We generate the synthetic storms using the multivariate sea-storm model from Wahl et al. (2016), which identifies interdependencies among the most relevant sea-storm variables using a vine copula model with the 35-year empirical storm record as input (Text S1).

2.2. Dunes

The elevation of a dune cell (Z_d) is taken as the sum of the dune toe elevation (Z_t), which remains fixed over time relative to sea level, and the height of the dune crest (H_d) above its toe: $Z_d = Z_t + H_d$. At the beginning of each modeled year, dune cells grow in a logistic manner following Houser et al. (2015) up to a maximum dune height ($H_{d_{max}}$; Durán & Moore, 2013):

$$\Delta H_d = r H_d \left(1 - \frac{H_d}{H_{d_{\rm max}}} \right)$$

where *r* is the growth rate controlling the shape of the logistic curve. Within the model, the dune growth rate (*r*) varies randomly for each dune cell alongshore between a user-defined r_{\min} and r_{\max} , following observations from Houser et al. (2015). The characteristic dune growth rate (\overline{r}) is calculated as the mean of r_{\min} and r_{\max} . When varying r_{\min} and r_{\max} , we hold the range between the two values constant at 0.5 (Houser et al., 2015). Additionally, the static width of the dune domain (W_d) determines the number of rows (cross-shore) that make up the active dune field, with ΔH_d decaying exponentially for each row in the landward direction.

Where water elevations exceed dune elevations during storms, dune heights are reduced using a predictor of dune erosion as a function of total water level developed by Goldstein and Moore (2016). This predictor was created using data from Long et al. (2014) passed through a machine learning technique to yield a smooth nonlinear function relating the change in dune elevation (ΔZ_d) to the elevation of water above the prestorm dune elevation (i.e., R_{high}), both normalized by the pre-storm dune elevation:





Barrier3D does not simulate dune recovery during interstorm periods that fall within one model year, assuming that most dune growth likely occurs outside the peak storm season when the density of dune vegetation is higher and disturbance is relatively rare (e.g., Delgado-Fernandez & Davidson-Arnott, 2011; Montreuil et al., 2013; Ruz & Meur-Ferec, 2004; van Puijenbroek et al., 2017).

When the ocean shoreline erodes one cell length, the front row of the dune domain is removed and the first row of the barrier interior functionally becomes the back row of the active dune field (thereby maintaining W_d). When the ocean shoreline progrades one cell width, the most landward dune row functionally becomes part of the barrier interior (while retaining its elevation) and a new row is added to the front of the dune domain, with dune cell elevation set to Z_t .

2.3. Overwash

To simulate overwash processes landward of the dune crest, Barrier3D uses a version of the cellular flow routing scheme developed by Murray and Paola (1994, 1997), in which water introduced at overtopped dune cells is carried landward row-by-row over the barrier interior, transporting sediment with it (Text S1). When all the water has been routed over the barrier interior, the model updates the barrier elevation according to the difference between the volume of sediment lost and gained from each cell. This process occurs iteratively for each hour of storm duration.

Sediment transported across the barrier interior to the back-barrier is distributed across the bay following exponential decay and contributes to landward extension of the back-barrier shoreline. The bottom of the back-barrier bay is flat (except where coarse, immobile overwash sediment has been deposited) and set to a constant depth (here 3 m), following the assumption that depositional and erosional processes for fine back-barrier sediment are able to maintain an equilibrium depth over the course of the simulation (Marani et al., 2007).

2.4. Shoreline Change

Shoreline change, applied uniformly across the barrier segment, follows the equations of Lorenzo-Trueba and Ashton (2014; Text S1). Shoreline erosion or accretion results from a combination of: (a) RSLR; (b) the cumulative volume of sediment removed from the upper shoreface by overwash and dune growth; and (c) net sediment exchange between the upper and lower shoreface. Shoreface sediment flux depends on the shoreface slope, which tends to dynamically adjust towards an equilibrium in response to perturbations (i.e., overwash and dune growth).

3. Experiments and Results

By examining barrier behavior across wide ranges of several key input parameter values, our simulations investigate barrier dynamics relevant to most barrier systems. To provide a common starting point for all simulations and to ground the model in reality, we parameterize the initial morphology and storm climatology for Hog Island within the Virginia Coast Reserve (VCR), VA, USA (Figure 1a). The initial morphology for the interior domain in our experiments consists of a digital elevation model (DEM) for a 500-m-long segment of Hog Island discretized into 10-by-10 m cells. $H_{d_{max}}$ and Z_t are based on values extracted from a Hog Island DEM, and the synthetic storms are derived from wave hindcast data offshore Hog Island (USACE Wave Information Studies) and the nearest tide gauge in Wachapreague, VA. For a list of all input variables and the values used in these experiments, see Table S1.

Each simulation runs for 1,000 years or until the barrier drowns. Punctuated retreat is determined to occur if the simulation includes two or more alternating periods of both transgression and relative immobility;



Geophysical Research Letters



Figure 2. Dune-driven punctuated retreat. (a) Probability (n = 100) of punctuated retreat across a range of storm frequencies (\overline{N}_s) and characteristic dune growth rates (\overline{r}). Average duration of (b) immobile and (c) transgressive periods across the same \overline{N}_s - \overline{r} parameter space. (d–f) Probability density functions (PDFs) of average annual dune height at three different locations within the parameter space for periods of immobility (green) and transgression (red); locations in parameter space are indicated by the corresponding lettering within (a–c).

otherwise, the behavior is considered continuous. Periods of immobility (transgression) are defined as 30 years or more of shoreline change rates (SCR) under (over) 0.5 m/yr (see Text S1 for additional details). Varying the duration and SCR thresholds used to define punctuated retreat produces predictable differences in our results but does not change the fundamental conclusions we draw from them (Figures S1 and S2).

3.1. Dune-Driven Discontinuous Retreat

To examine how dune-storm interactions influence barrier migration, we run simulations across a range of storm frequencies, given by the average number of storms per year (\overline{N}_s), and characteristic dune growth rates (\overline{r}) (Figure 2). RSLR is held at 4 mm/yr. To account for storm stochasticity within each unique simulation, we run each of these parameter combinations 100 times and report the mean of these 100 simulations for our results.

We find that punctuated retreat is most probable diagonally across the center of the parameter space, where the storm return period and characteristic time scale of dune growth are of similar magnitudes (Figure 2a). Here, dunes are more likely to exhibit bistable behavior (Figure 2d), transitioning between stable states of tall, overwash-resistant dunes and low, overwash-vulnerable dunes. This dune behavior is consistent with previous models of dune-storm interactions (Durán Vinent & Moore, 2015; Durán Vinent et al., 2021; Gold-stein & Moore, 2016). The bistability of dunes directly impacts overwash flux and, consequently, barrier migration, evidenced by instances of taller dunes occurring predominantly within periods of immobility and shorter dunes within periods of transgression (Figure 2d). Additionally, the average durations of immobile and transgressive periods are of similar magnitude where dunes are bistable (Figures 2b and 2c), suggesting that the barriers tend to spend relatively equal amounts of time stationary as actively retreating. During periods of immobility, when overwash fluxes into the barrier interior are limited, the barrier tends to lose elevation relative to sea level and become narrower from passive inundation of the back-barrier shoreline; the barrier tends to aggrade and widen during periods of transgression (Figure S3).

When the storm return period and characteristic dune-growth time scale are of dissimilar magnitudes, punctuated retreat is less likely. The combination of fast dune growth and low storm frequency tends to result in the formation of tall dunes that limit overwash flux and reform quickly if knocked down (Figure 2e). This typically precludes periods of barrier transgression during the simulations in favor of continuous immobility and, ultimately, passive drowning via RSLR. As such, transgressive periods tend to be considerably shorter-lived than immobile periods (Figures 2b and 2c). Conversely, the combination of slow dune growth and high storm frequency tends to prevent dunes from ever recovering (Figure 2f), resulting in continuous barrier transgression. Therefore, the average duration of transgressive periods tends to be much longer than the average duration of immobile periods (Figures 2b and 2c). Punctuated retreat can occur within these corners of the parameter space (Figure 2a), sometimes as the result of anomalous periods of high or low storm activity or potentially time lags in the shoreface response to overwash (Figure S4).

3.2. Sea-Level Rise

Next, we explore the impacts of RSLR on discontinuous retreat by running the same 25 combinations of $\overline{N_s}$ and \overline{r} presented in Section 3.1 (RSLR = 4 mm/yr) at two additional RSLR rates: 2 and 8 mm/yr (Figure 3). Overall, the likelihood of discontinuous barrier migration decreases with higher rates of RSLR (Figures 3a–3c). Additionally, periods of transgression (Figures 3g–3i) tend to be much longer than periods of immobility (Figures 3d–3f) at higher rates of RSLR.

These results stem in part from the model implementation of the Bruun (1962) rule. Greater background shoreline retreat rates resulting from higher rates of RSLR make it less likely for periods of immobility to occur even in the absence of significant overwash. However, increased background shoreline retreat rates alone do not fully explain the trends we observe (Figure S5). Higher rates of RSLR also typically lead to shorter dunes (Figure S6), the result of increased landward migration of the dune crest in response to faster shoreline change rates (Keijsers et al., 2016; Rodriguez et al., 2018), and reduced interior elevations, which increases accommodation space for overwash. Both of these effects allow for greater overwash fluxes, and therefore greater barrier migration rates, tending to reduce the likelihood of immobile periods and therefore punctuated retreat in favor of continuous barrier transgression and, if sea level rises faster than overwash fluxes can maintain the barrier elevation, barrier drowning (though to maintain our focus on migration behaviors influenced by dune dynamics, we do not explore high enough RSLR rates in our simulations to produce this particular drowning behavior).

3.3. Storm Intensity

To explore the impacts of changing storm intensity, we run the same 25 combinations of \overline{N}_s and \overline{r} presented in Section 3.1 with both higher and lower storm intensities (Figure 4), with storm intensity represented by the TWL. Instead of drawing randomly from the 10,000 simulated storms as before, we bin them by R_{high} , fit a beta distribution to the histogram, and draw from the distribution shifted +0.15 (-0.15) m such that bins containing storms with higher (lower) R_{high} values are preferentially chosen; storms are then selected randomly from the specified bin. We refer to the three levels of storm intensity as "low," "observed" (i.e., the same simulations presented in Figure 2), and "high."

Increasing storm intensity shifts the maximum potential for discontinuous migration towards barriers under conditions of lower storm frequency (Figures 4a–4c). Dune bistability (and therefore discontinuous retreat) becomes more likely at lower storm frequencies because a greater proportion of storms overtop the dunes when storms are more intense. Barriers that typically experience continuous immobility as a result of tall, resilient dunes at low storm intensity transition to discontinuous retreat at high storm intensity as dunes become bistable; barriers that typically exhibit punctuated retreat at low storm intensity transition to continuous transgression at high storm intensity as low-elevation dunes become the single attracting state of the dune system. The same shifts occur for the average durations of immobile and transgressive periods (Figures 4d–4i), as more barriers experience longer transgressive periods and shorter immobile periods with higher storm intensities.

Geophysical Research Letters





Figure 3. Varying relative sea-level rise (RSLR). (a–c) Probability of punctuated retreat, (d–f) average duration of immobile periods, and (g–i) average duration of transgressive periods, across the same \overline{N}_{s} - \overline{r} parameter space at three different rates of RSLR. Panels b, e, and h are identical to Figures 2a–2c. Diagonal lines

4. Discussion and Conclusions

The goal of this work is to explore and explain large-scale barrier behavior and the key feedbacks that give rise to it, rather than to numerically predict the evolution of any particular location or environment. The standard exploratory modeling approach we adopt here - reducing the number of processes involved and treating them in considerably simplified ways to enhance the clarity of potential insights - supports this goal, but bears several limitations worth considering. For instance, we do not model a beach domain, and instead assume that beach characteristics remain invariant over time and space. In reality, beach characteristics (e.g., slope and width) can vary greatly over hourly to decadal time scales (e.g., O'Dea et al., 2019) and 10²-10³ m length scales (Vos et al., 2020) in ways that can affect dune growth (Durán & Moore, 2013), dune erosion (Beuzen et al., 2019), and overwash (Donnelly et al., 2006). Additionally, the model does not include tidal inlet processes, which can account for a significant proportion of overall transgressive flux in a barrier system (e.g., Leatherman, 1979; Nienhuis & Lorenzo-Trueba, 2019). The current model formulation also lacks some of the sediment pathways that can contribute to deflation of the barrier (i.e., narrowing and flattening; Passeri et al., 2020), such as gradients in alongshore transport, breaching, or storm-driven seaward transport, which would all tend to increase the vulnerability of the barrier to drowning in our model.

Nevertheless, although simplifications may limit the quantitative precision of our results, our model experiments provide valuable qualitative insights into barrier behavior and how it may change in the future. We

indicate areas of no data.



Geophysical Research Letters



Figure 4. Varying storm intensity. (a–c) Probability of punctuated retreat, (d–f) average duration of immobile periods, and (g–i) average duration of transgressive periods, across the same $\overline{N}_s \cdot \overline{r}$ parameter space at three different levels of storm intensity. Panels b, e, and h are identical to Figures 2a–2c.

find that punctuated retreat is a common behavioral regime, consistent with recent modeling (Ashton & Lorenzo-Trueba, 2018; Ciarletta et al., 2019) and observational (e.g., Mellett & Plater, 2018; Raff et al., 2018) studies. In contrast to previous work, however, the discontinuous retreat in our experiments is driven principally by dune dynamics, though this does not invalidate other explanations for discontinuous behavior or suggest that multiple mechanisms cannot operate in conjunction. In fact, intrinsic couplings between dune dynamics and other proposed forcing mechanisms can alter the potential for dune state change. For example, RSLR can increase the vulnerability of dunes to erosion and therefore overwash (a dynamic we capture in the model; see also Rodriguez et al., 2018), and changes in sediment supplied to beaches (e.g., from the shoreface) can limit or bolster dune recovery (Houser, Barrineau, et al., 2018). In the discontinuous retreat of our model, dune dynamics lead to interior elevation loss (relative to MSL) during periods of immobility. This contrasts with the shoreface-driven discontinuous retreat from the model of Lorenzo-Trueba and Ashton (2014), in which periods of immobility are characterized by barrier aggradation (i.e., the barrier maintains elevation relative to sea level; see Text S2 for a comprehensive comparison of discontinuous retreat in the two models).

In response to projected accelerated RSLR, our experiments suggest that dune-driven discontinuous retreat will become less common overall in favor of continuous transgression as immobile periods become shorter and more infrequent. Meanwhile, increasing storm intensity will shift the range of storm frequencies under which punctuated retreat is most likely to occur, becoming the dominant behavior of barriers that were

previously stable. Barrier migration behaviors could vary latitudinally, as a result of latitudinal gradients in storm characteristics and, conceivably, dune growth rates. Therefore, projected climate-driven shifts in maximum storm intensity (e.g., Knutson et al., 2020) and the distribution of dune grass species (e.g., Gold-stein et al., 2018) may result in latitudinal shifts in barrier behavior; these are potentially fruitful topics for future research.

Dune-driven punctuated retreat is supported by empirical evidence from Durán Vinent and Moore, (2015) relating lower dune heights to faster shoreline change rates from the VCR. Houser, Wernette, et al. (2018) observe a similar association between dune height and storm-scale (<10 years) shoreline change from barrier islands in Texas and Florida, but raise important questions about length scales given that a single barrier can exhibit km-scale reaches of both high and low dunes (Zinnert et al., 2016). On barriers exhibiting km-scale alongshore variability in dune state, alongshore transport would tend to redistribute sediment from sections with tall dunes towards overwashing areas without dunes (Houser, Wernette, et al., 2018), tempering local effects of dune-driven punctuated retreat. The extent of punctuated behavior thus depends on the km-scale alongshore continuity of dune state, as well as the strength of alongshore diffusion of shoreline change rates.

These results demonstrate the importance of internal controls and nonlinear state changes in barrier evolution. Barrier rollover is commonly perceived (and modeled) as a constant background process, but over time scales of even a few decades to several centuries, our results suggest that is not always the case. For modeling barriers on time scales relevant to coastal communities and managers, we find that explicitly including dune dynamics - erosion, recovery, and interaction with storm overwash processes - is essential in determining both the rate and style of barrier retreat. Further, dune-driven nonlinearity in barrier retreat raises questions about the reliability of linear regression for calculating long-term historical shoreline change, which might mischaracterize the historical behavior or perceived stability of a barrier. Without contextualization for dune-storm dynamics, such historical observations could fail to provide insights necessary for accurately forecasting coastal change. Future work is needed to address the potential for cascading state changes across the entire barrier-marsh-bay system triggered by dune-driven discontinuous barrier retreat.

Data Availability Statement

Barrier3D is available for download from the online Community Surface Dynamics Modeling System model repository at https://csdms.colorado.edu/wiki/Model:Barrier3D. Data for model experiments are in repository at https://doi.org/10.6073/pasta/d2b133e539f87f38f241b97e64ddaa06.

References

- Ashton, A. D., & Lorenzo-Trueba, J. (2018). Morphodynamics of barrier response to sea-level rise. In L. J. Moore, & A. B. Murray (Eds.), *Barrier dynamics and response to changing climate* (pp. 277–304). Springer International Publishing. https://doi.org/10.1007/978-3-319-68086-610.1007/978-3-319-68086-6_9
- Beuzen, T., Harley, M. D., Splinter, K. D., & Turner, I. L. (2019). Controls of variability in berm and dune storm erosion. Journal of Geophysical Research: Earth Surface, 124, 2647–2665. https://doi.org/10.1029/2019JF005184
- Bruun, P. (1962). Sea-level rise as a cause of shore erosion. Journal of the Waterways and Harbors Division, 88, 117–130. https://doi.org/10.1061/jwheau.0000252

Ciarletta, D. J., Lorenzo-Trueba, J., & Ashton, A. D. (2019). Mechanism for retreating barriers to autogenically form periodic deposits on continental shelves. *Geology*, 47, 239–242. https://doi.org/10.1130/g45519.1

Cooper, J. A. G., Green, A. N., Meireles, R. P., Klein, A. H. F., Souza, J., & Toldo, E. E. (2016). Sandy barrier overstepping and preservation linked to rapid sea level rise and geological setting. *Marine Geology*, 382, 80–91. https://doi.org/10.1016/j.margeo.2016.10.003

Delgado-Fernandez, I., & Davidson-Arnott, R. (2011). Meso-scale aeolian sediment input to coastal dunes: The nature of aeolian transport events. *Geomorphology*, 126, 217–232. https://doi.org/10.1016/j.geomorph.2010.11.005

Dolan, R., & Godfrey, P. (1973). Effects of Hurricane Ginger on the barrier islands of North Carolina. *Geological Society of America Bulletin*, 84, 1329–1334. https://doi.org/10.1130/0016-7606(1973)84<1329:EOHGOT>2.0.CO;2

Donnelly, C., Kraus, N., & Larson, M. (2006). State of knowledge on measurement and modeling of coastal overwash. *Journal of Coastal Research*, 22, 965–991. https://doi.org/10.2112/04-0431.1

Acknowledgments

This work is supported by the Virginia Coast Reserve Long-Term Ecological Research Program (National Science Foundation DEB-1832221), the NSF Coupled Natural Human Dynamics Program (CNH-L-1715638), and the Preston Jones and Mary Elizabeth Frances Dean Martin Fellowship Fund from the Department of Geological Sciences at the University of North Carolina at Chapel Hill. The authors thank Thomas Wahl for guidance with the storm model, and acknowledge computing time on the CU-CSDMS High-Performance Computing Cluster.

Durán, O., & Moore, L. J. (2013). Vegetation controls on the maximum size of coastal dunes. Proceedings of the National Academy of Sciences of the USA, 110, 17217–17222. https://doi.org/10.1073/pnas.1307580110

Durán Vinent, O., & Moore, L. J. (2015). Barrier island bistability induced by biophysical interactions. *Nature Climate Change*, 5, 1758–6798. https://doi.org/10.1038/nclimate2474

Durán Vinent, O., Schaffer, B. E., & Rodriguez-Iturbe, I. (2021). Stochastic dynamics of barrier island elevation. *Proceedings of the National Academy of Sciences of the USA*, 118, e2013349118. https://doi.org/10.1073/pnas.2013349118

- Forbes, D. L., Orford, J. D., Carter, R. W. G., Shaw, J., & Jennings, S. C. (1995). Morphodynamic evolution, self-organisation, and instability of coarse-clastic barriers on paraglacial coasts. *Marine Geology*, 126, 63–85. https://doi.org/10.1016/0025-3227(95)00066-8
- Forbes, D. L., Taylor, R. B., Orford, J. D., Carter, R. W. G., & Shaw, J. (1991). Gravel-barrier migration and overstepping. *Marine Geology*, 97, 305–313. https://doi.org/10.1016/0025-3227(91)90122-K
- Goldstein, E. B., & Moore, L. J. (2016). Stability and bistability in a one-dimensional model of coastal foredune height. *Journal of Geophysical Research: Earth Surface*, 121, 964–977. https://doi.org/10.1002/2015JF003783
- Goldstein, E. B., Mullins, E. V., Moore, L. J., Biel, R. G., Brown, J. K., Hacker, S. D., et al. (2018). Literature-based latitudinal distribution and possible range shifts of two US east coast dune grass species (Uniola paniculata and Ammophila breviligulata). PeerJ, 6, e4932. https://doi.org/10.7717/peerj.4932
- Houser, C., Barrineau, P., Hammond, B., Saari, B., Rentschler, E., Trimble, S., et al. (2018a). Role of the foredune in controlling barrier island response to sea level rise. In L. J. Moore, & A. B. Murray (Eds.), *Barrier dynamics and response to changing climate* (pp. 175–207). Springer International Publishing. https://doi.org/10.1007/978-3-319-68086-610.1007/978-3-319-68086-6_6
- Houser, C., Wernette, P., Rentschlar, E., Jones, H., Hammond, B., & Trimble, S. (2015). Post-storm beach and dune recovery: Implications for barrier island resilience. *Geomorphology*, 234, 54–63. https://doi.org/10.1016/j.geomorph.2014.12.044
- Houser, C., Wernette, P., & Weymer, B. A. (2018b). Scale-dependent behavior of the foredune: Implications for barrier island response to storms and sea-level rise. *Geomorphology*, 303, 362–374. https://doi.org/10.1016/j.geomorph.2017.12.011
- Keijsers, J. G. S., Groot, A. V. D., & Riksen, M. J. P. M. (2016). Modeling the biogeomorphic evolution of coastal dunes in response to climate change. Journal of Geophysical Research: Earth Surface, 121, 1161–1181. https://doi.org/10.1002/2015JF003815
- Knutson, T., Camargo, S. J., Chan, J. C., Emanuel, K., Ho, C. H., Kossin, J., et al. (2020). Tropical cyclones and climate change assessment: Part II: Projected response to anthropogenic warming. *Bulletin of the American Meteorological Society*, 101, E303–E322. https://doi. org/10.1175/BAMS-D-18-0194.1
- Leatherman, S. P. (1979). Migration of Assateague Island, Maryland, by inlet and overwash processes. *Geology*, 7, 104–107. https://doi. org/10.1130/0091-7613(1979)7<104:MOAIMB>2.0.CO;2
- Long, J. W., de Bakker, A. T. M., & Plant, N. G. (2014). Scaling coastal dune elevation changes across storm-impact regimes. *Geophysical Research Letters*, 41, 2899–2906. https://doi.org/10.1002/2014GL059616
- Lorenzo-Trueba, J., & Ashton, A. (2014). Rollover, drowning, and discontinuous retreat: Distinct modes of barrier response to sea-level rise arising from a simple morphodynamic model. *Journal of Geophysical Research: Earth Surface*, 119, 779-801. https://doi. org/10.1002/2013JF002941
- Marani, M., D'Alpaos, A., Lanzoni, S., Carniello, L., & Rinaldo, A. (2007). Biologically-controlled multiple equilibria of tidal landforms and the fate of the Venice lagoon. *Geophysical Research Letters*, 34. https://doi.org/10.1029/2007GL030178
- Mellett, C. L., Hodgson, D. M., Lang, A., Mauz, B., Selby, I., & Plater, A. J. (2012). Preservation of a drowned gravel barrier complex: A landscape evolution study from the north-eastern English Channel. *Marine Geology*, *315–318*, 115–131. https://doi.org/10.1016/j. margeo.2012.04.008
- Mellett, C. L., & Plater, A. J. (2018). Drowned barriers as archives of coastal-response to sea-level rise. In L. J. Moore, & A. B. Murray (Eds.), *Barrier dynamics and response to changing climate* (pp. 57–89). Springer International Publishing. https://doi.org/10.1007/978-3-319-68086-610.1007/978-3-319-68086-6_2
- Montreuil, A.-L., Bullard, J. E., Chandler, J. H., & Millett, J. (2013). Decadal and seasonal development of embryo dunes on an accreting macrotidal beach: North Lincolnshire, UK. Earth Surface Processes and Landforms, 38, 1851–1868. https://doi.org/10.1002/esp.3432
- Moore, L. J., List, J. H., Williams, S. J., & Stolper, D. (2010). Complexities in barrier island response to sea level rise: Insights from numerical model experiments, North Carolina Outer Banks. *Journal of Geophysical Research*, 115, F03004. https://doi.org/10.1029/2009JF001299
- Murray, A. B. (2003). Contrasting the goals, strategies, and predictions associated with simplified numerical models and detailed simulations. In *Prediction in geomorphology* (pp. 151–176). Washington D.C.:AGU. https://doi.org/10.13008/2153-3695.1710
- Murray, A. B. (2013). Which models are good (enough), and when? In J. F. Shroder (Ed.), *Treatise on geomorphology* (pp. 50–58). Elsevier: Academic Press. https://doi.org/10.1016/b978-0-12-374739-6.00027-0
- Murray, A. B., & Paola, C. (1994). A cellular model of braided rivers. Nature, 371, 54-57. https://doi.org/10.1038/371054a0
- Murray, A. B., & Paola, C. (1997). Properties of a cellular braided-stream model. *Earth Surface Processes and Landforms*, 22, 1001–1025. https://doi.org/10.1002/(SICI)1096-9837(199711)2210.1002/(sici)1096-9837(199711)22:11<1001::aid-esp798>3.0.co;2-0
- Nienhuis, J. H., & Lorenzo-Trueba, J. (2019). Can barrier islands survive sea-level rise? Quantifying the relative role of tidal inlets and overwash deposition. *Geophysical Research Letters*, 46, 14613–14621. https://doi.org/10.1029/2019GL085524
- Nordfjord, S., Goff, J. A., Austin, J. A., Jr, & Duncan, L. S. (2009). Shallow stratigraphy and complex transgressive ravinement on the New Jersey middle and outer continental shelf. *Marine Geology*, 266, 232–243. https://doi.org/10.1016/j.margeo10.1016/j.margeo.2009.08.010
- O'Dea, A., Brodie, K. L., & Hartzell, P. (2019). Continuous coastal monitoring with an automated terrestrial lidar scanner. Journal of Marine Science and Engineering, 7, 37. https://doi.org/10.3390/jmse7020037
- Passeri, D. L., Dalyander, P. S., Long, J. W., Mickey, R. C., Jenkins, R. L., III, Thompson, D. M., et al (2020). The roles of storminess and sea level rise in decadal barrier island evolution. *Geophysical Research Letters*, 47, e2020GL089370. https://doi.org/10.1029/2020GL089370
- Raff, J. L., Shawler, J. L., Ciarletta, D. J., Hein, E. A., Lorenzo-Trueba, J., & Hein, C. J. (2018). Insights into barrier-island stability derived from transgressive/regressive state changes of Parramore Island, Virginia. *Marine Geology*, 403, 1–19. https://doi.org/10.1016/j. margeo.2018.04.007
- Rampino, M. R., & Sanders, J. E. (1982). Holocene transgression in south-central Long Island, New York Reply. Journal of Sedimentary Research, 52, 1020–1025.
- Rodriguez, A. B., Yu, W., & Theuerkauf, E. J. (2018). Abrupt increase in washover deposition along a transgressive barrier island during the late nineteenth century acceleration in sea-level rise. In L. J. Moore, & A. B. Murray (Eds.), Barrier dynamics and response to changing climate (pp. 121–145). Springer International Publishing. https://doi.org/10.1007/978-3-319-68086-6_4
- Ruz, M. H., & Meur-Ferec, C. (2004). Influence of high water levels on aeolian sand transport: Upper beach/dune evolution on a macrotidal coast, Wissant Bay, northern France. *Geomorphology*, 60, 73–87. https://doi.org/10.1016/j.geomorph.2003.07.011
- Sallenger, A. H. (2000). Storm impact scale for barrier islands. Journal of Coastal Research, 16, 890-895.
- Storms, J. E. A., Weltje, G. J., Terra, G. J., Cattaneo, A., & Trincardi, F. (2008). Coastal dynamics under conditions of rapid sea-level rise: Late Pleistocene to Early Holocene evolution of barrier-lagoon systems on the northern Adriatic shelf (Italy). Quaternary Science Reviews, 27, 1107–1123. https://doi.org/10.1016/j.quascirev.2008.02.009
- Stutz, M. L., & Pilkey, O. H. (2011). Open-ocean barrier islands: Global influence of climatic, oceanographic, and depositional settings. Journal of Coastal Research, 27, 207–222. https://doi.org/10.2112/09-1190.1



- Sweet, W. V., Kopp, R. E., Weaver, C. P., Obeysekera, J., Horton, R., Thieler, E. R., & Zervas, C. (2017). Global and regional sea level rise scenarios for the United States (Technical Report NOS CO-OPS 083). Silver Spring, MD: National Oceanic and Atmospheric Administration. van Puijenbroek, M. E. B., Limpens, J., de Groot, A. V., Riksen, M., Gleichman, M., Slim, P. A., et al (2017). Embryo dune development
- drivers: Beach morphology, growing season precipitation, and storms. *Earth Surface Processes and Landforms*, 42, 1733–1744. https://doi.org/10.1002/esp.4144
 Vos, K., Harley, M. D., Splinter, K. D., Walker, A., & Turner, I. L. (2020). Beach slopes from satellite-derived shorelines. *Geophysical Re-*
- search Letters, 47, e2020GL088365. https://doi.org/10.1029/2020GL088365
- Wahl, T., Plant, N. G., & Long, J. W. (2016). Probabilistic assessment of erosion and flooding risk in the northern Gulf of Mexico. Journal of Geophysical Research: Oceans, 121, 3029–3043. https://doi.org/10.1002/2015JC011482
- Zinnert, J. C., Brantley, S. T., & Young, D. R. (2016). Bistability and the future of barrier islands. *Nature Climate Change*, 6, 5–6. https://doi.org/10.1038/nclimate2802

References From the Supporting Information

- Daneshkhah, A., Remesan, R., Chatrabgoun, O., & Holman, I. P. (2016). Probabilistic modeling of flood characterizations with parametric and minimum information pair-copula model. *Journal of Hydrology*, 540, 469–487. https://doi.org/10.1016/j.jhydrol.2016.06.044
- Jane, R., Cadavid, L., Obeysekera, J., & Wahl, T. (2020). Multivariate statistical modelling of the drivers of compound flood events in south Florida. *Natural Hazards and Earth System Sciences*, 20, 2681–2699. https://doi.org/10.5194/nhess-20-2681-2020
- Larson, M., Wise, R. A., & Kraus, N. C. (2004). Coastal overwash: Part 2, upgrade to SBEACH (technical report ERDC/RSM-TN-15). Vicksburg, MS: U.S. Army Engineer Research and Development Center.
- Li, F., van Gelder, P. H. A. J. M., Vrijling, J. K., Callaghan, D. P., Jongejan, R. B., & Ranasinghe, R. (2014). Probabilistic estimation of coastal dune erosion and recession by statistical simulation of storm events. *Applied Ocean Research*, 47, 53–62. https://doi.org/10.1016/j. apor.2014.01.002
- Nagler, T., Schepsmeier, U., Stoeber, J., Brechmann, E. C., Graeler, B., Erhardt, T., et al (2020). *Package 'VineCopula'*. *R package version 2.4.1*. Retrieved from https://cran.r-project.org/web/packages/VineCopula/VineCopula.pdf
- Pawlowicz, R., Beardsley, B., & Lentz, S. (2002). Classical tidal harmonic analysis including error estimates in MATLAB using T_TIDE. Computers & Geosciences, 28, 929–937. https://doi.org/10.1016/S0098-3004(02)00013-4
- Sadegh, M., Ragno, E., & AghaKouchak, A. (2017). Multivariate copula analysis toolbox (MvCAT): Describing dependence and underlying uncertainty using a Bayesian framework. *Water Resources Research*, 53, 5166–5183. https://doi.org/10.1002/2016WR020242
- Schwartz, R. K. (1982). Bedform and stratification characteristics of some modern small-scale washover sand bodies. *Sedimentology*, *29*, 835–849. https://doi.org/10.1111/j.1365-3091.1982.tb00087.x
- Stockdon, H. F., Holman, R. A., Howd, P. A., & Sallenger, A. H., Jr. (2006). Empirical parameterization of setup, swash, and runup. *Coastal Engineering*, 53, 573–588. https://doi.org/10.1016/j.coastaleng.2005.12.005